Author's response to reviews

Title: Methods and Metrics Challenges of Delivery System Research

Authors:

Jeffrey A. Alexander (jalexand@umich.edu)
Larry R. Hearld (lhearld@uab.edu)

Version: 3 Date: 1 February 2012

Author's response to reviews:

Reviewer 2: "The authors accept that providing examples of effective interventions or studies in Table 1 is a useful one. However, instead of providing boxes or text, they have only put reference numbers. I actually think adding 5 short boxes would enhance the interest of the paper but even if this is included in Table 1 it is surely worth the modest effort of providing some text to illustrate the main points."

We misunderstood the reviewer’s original suggestion and appreciate the clarification. In Table 1 in the revised manuscript, we have inserted brief, narrative summaries of studies that illustrate how researchers have applied our recommendations for addressing the methodological issue/challenges presented in our paper. The changes are highlighted in the revised version of the manuscript.